Author's response to reviews

Title: Prevalence and severity of antipsychotic related constipation in patients with schizophrenia: a retrospective descriptive study

Authors:

Marc De Hert (marc.de.hert@uc-kortenberg.be)
Liesbeth Dockx (liesbeth.dockx@student.kuleuven.be)
Chiara Bernagie (chiara.bernagie@student.kuleuven.be)
Bie Peuskens (bie.peuskens@uc-kortenberg.be)
Kim Sweers (kim.sweers@uc-kortenberg.be)
Stefan Leucht (Stefan.Leucht@lrz.tu-muenchen.de)
Jan Tack (Jan.Tack@med.kuleuven.be)
Stefan Van de Straete (stefan.van.de.straete@uc-kortenberg.be)
Martien Wampers (martien.wampers@uc-kortenberg.be)
Joseph Peuskens (joseph.peuskens@uc-kortenberg.be)

Version: 4 Date: 5 March 2011

Author's response to reviews:

Formatting changes to be made:
-------------------------------

Major revisions (we require the author to make these changes)

References: please cite all references in numerical order in the text. Several citations appear out of sequence from reference 13 onwards.
changed

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Summary: rename Abstract and remove the Objectives subheading.
changed

Abstract: remove keywords.
done

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
done